Author’s response to reviews

Title: Clinical efficacy of submucosal injection of triamcinolone acetonide in the treatment of type II/III interstitial cystitis/bladder pain syndrome

Authors:

Tao Jiang (540126004@qq.com)
Xiaozhou Zhou (ai_touch13@sina.com)
Zhipeng Chen (306817794@qq.com)
Tailing Xiong (xtl1986@stu.xjtu.edu.cn)
Jian Fu (xingzhu007@163.com)
Zhengchao Liu (lzc1002@qq.com)
Dishi Yan (490800596@qq.com)
Zhansong Zhou (zzs68754186@163.com)
Wenhao Shen (chongqingswh@aliyun.com)

Version: 3 Date: 18 Feb 2020

Author’s response to reviews:

Dear Editor and reviewers,

We are glad to get the further instructions on the manuscript "Clinical efficacy of submucosal injection of triamcinolone acetonide in the treatment of type II/III interstitial cystitis/bladder pain syndrome" (BURO-D-19-00611R2). Thanks for your kindly suggestions. I revised the manuscript carefully according to your demands and we are looking forward to your positive response. Please find the answers to the questions in the following part.

Best wishes to you.

Yours Sincerely,

Wenhao Shen M.D.

Urology Institute of People Liberation Army, Southwest Hospital, Third Military Medical University (Army Medical University),
1. Consent to participate

In your “ethical approval and consent to participate” section, please confirm whether informed consent, obtained from all patients, was written or verbal and clearly state this in your manuscript. This should match the relevant statement provided in the methods section.

We have revised the “ethical approval and consent to participate” section.

“All patients agreed and signed the informed consent, that their information (including clinical information and follow-up data) would be collected for scientific study and by published in professional medical journals.”

Thank you!

2. Remove supplementary file

Please remove the supplementary “Response letter-R2.doc” from the manuscript, as these files are not required to be attached to the manuscript at this stage of the editorial process.

We have removed the supplementary “Response letter-R2.doc” from the manuscript. Thank you!

3. Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

We have uploaded the manuscript as a single, final, clean version.

Thank you!